Author's response to reviews

Title: Acetylcholinesterase Inhibitors (AChEi's) for the treatment of visual hallucinations in schizophrenia: A case report.

Authors:

Sachin S Patel (sachin.patel@slam.nhs.uk)
Azizah Attard (azizah.attard@slam.nhs.uk)
Pamela Jacobsen (pamela.jacobsen@slam.nhs.uk)
Sukhi Shergill (sukhi.shergill@kcl.ac.uk)

Version: 2 Date: 26 August 2010

Author's response to reviews:

Dear Sir/ Madam,

We have revised the manuscript as requested.

1. The referencing has been changed from superscript to boxed
2. The title page has been edited in line with policies- including the country details
3. We have also reformatted to remove underlined headings throughout.

Regards

Dr Sukhi Shergill
26.8.10